Author’s response to reviews

Title: Pattern and determinants of contraceptive usage among women of reproductive age from the Digo community residing in Kwale, Kenya: Results from a cross-sectional household survey

Authors:

Vernon Mochache (vmochache@yahoo.com)
Amyn Lakhani (amyn.lakhani@aku.edu)
Hajara El-Busaidy (hajarabusaidy13@gmail.com)
Marleen Temmerman (marleen.temmerman@aku.edu)
Peter Gichangi (peter@icrhk.org)

Version: 4 Date: 27 Nov 2017

Author’s response to reviews:

Dear Editor,

Thank you for reviewing our manuscript and getting it closer to publication. Below is a point-by-point response on the editorial comments:

1) Please add individual email addresses for all authors to the title page.

Response: We have included email addresses for all listed authors in the title page.

2) Consent to publish: this refers to consent obtained from patients to publish their individual personal information. Since there is no individual, non-anonymized information reported in your manuscript, this should include a "Not applicable" statement.

Response: Thank you for the clarification. We have edited the 'Consent to Publish' declaration to include a 'Not applicable' statement.

We look forward to further feedback on the manuscript.

Regards,

Dr. Vernon Mochache